Author's response to reviews

Title: Changes in public attitudes towards confidential adolescent sexual and reproductive health services in Lithuania after the introduction of new legislation: findings of the cross-sectional surveys conducted in 2005 and 2012

Authors:

Lina Jaruseviciene (ljaruseviciene@gmail.com)
Apolinaras Zaborskis (Apolinaras.Zaborskis@lsmuni.lt)
Skirmante Sauliune (smante@centras.lt)
Gediminas Jarusevicius (gedijaru@yahoo.com)
Jeffrey V Lazarus (jla@cphiv.dk)

Version: 2 Date: 6 July 2014

Author's response to reviews:

Dear Editor,

as is was suggested we have revised our manuscript and included line and page numbers.

We hope that now our manuscript is eligible for the peer review.

On behalf of authors,

Lina Jaruseviciene